Author's response to reviews

Title: Reduction of novel circulating long-chain fatty acids and colorectal cancer patients is independent of tumor burden and correlates with age.

Authors:

Shawn A Ritchie (s.ritchie@phenomenome.com)
Doug Heath (d.heath@phenomenome.com)
Yasuyo Yamazaki (y.yamazaki@phenomenome.com)
Bryan Grimmalt (b.grimmalt@phenomenome.com)
Amir Kavianpour (a.kavianpour@phenomenome.com)
Kevin Krenitsky (kkrenitsky@enzo.com)
Hoda Elshoni (hae571@mail.usask.ca)
Ichiro Takemasa (itakemasa@gesurg.med.osaka-u.ac.jp)
Masakazu Miyake (m.masakazu@k4.dion.ne.jp)
Mitsugu Sekimoto (msekimoto@gesurg.med.osaka-u.ac.jp)
Morito Monden (monden@hq.osaka-u.ac.jp)
Takeshi Tomonaga (tomonaga@nibio.go.jp)
Hisahiro Matsubara (matsuhm@faculty.chiba-u.jp)
Kazuyuki Sogawa (sgw1124jp@yahoo.co.jp)
Kazuyuki Matsushita (kmatsu@faculty.chiba-u.jp)
Fumio Nomura (fnomura@faculty.chiba-u.jp)
Dayan B Goodenowe (d.goodenowe@phenomenome.com)

Version: 4 Date: 29 November 2010

Author's response to reviews: see over
Dear BMC Editorial Production Team,

Listed below are the requested changes which have been made to the manuscript:

1. All tracked changes from previous version have been removed
2. The statement “All authors have read and approved the final version” has been added to the end of the author contributions section.
3. The “Competing Interests” section has been inserted directly before author contributions
4. There should be no full blank pages in the document other than blank space following the end of a section.
5. Reference #22 has been fixed.
6. The use of “Figure” has been consistently used throughout the manuscript. The only changes made were in the Figure legends where the titles were previously named Fig, and are now changed to “Figure”.

Sincerely,

Shawn Ritchie, corresponding author.